%Latex Version: TeX, C Version 3.14t3
%dvi2ps Version: dvips 5.521 Copyright 1986, 1993 Radical Eye Software

%\documentstyle[twocolumn,psfig{article}
\documentstyle[psfig,12pt]{article}

%\pagestyle{empty}

\setlength{\textwidth}{6.9in}
\setlength{\textheight}{8.75in}
\setlength{\columnsep}{2.5pc}
\setlength{\topmargin}{-0.3in}
\setlength{\oddsidemargin}{-.3in}
\setlength{\parindent}{1pc}
  
\def\lang{$<$}
\def\rang{$>$}
\newcommand{\DS}{\renewcommand{\baselinestretch}{1.2}
                 \tiny \normalsize}
\newcommand{\SS}{\renewcommand{\baselinestretch}{1}
                 \tiny \normalsize}


\begin{document}
\begin{thebibliography}{99}

\bibitem{RTLibs} I.~Ahmed, et al.,
{\bf Implementation and Scalability of Fortran 90D Intrinsic Functions on
Distributed Memory Machines,}
Technical Report SCCS-256, Northeast Parallel Architectures Center.

\bibitem{Bozkus} Z.~Bozkus, et al.,
{\bf Compiling Fortran 90D/HPF for Distributed Memory MIMD Computers,}
J.~of Parallel and Distributed Computing, (21)1, April 1994.

\bibitem{Unify} F-C.~Cheng, P.~Vaughan, D.~Reese, A.~Skjellum,
{\bf The Unify System,} Technical Report,
NSF Engineering Research Center, Mississippi State University,
September 1994.

\bibitem{Cowie} J.~Cowie, K.~Dincer, and X.~Li,
{\bf Towards a Web-based PCRC Programming Environment,}
Internal Technical Report, Northeast Parallel Architectures Center, Dec.
1995.

\bibitem{PVM} A.~Geist, A.~Beguelin, J.~Dongarra, W.~Jiang, R.~Manchek,
and V.~Sunderam,
{\bf PVM 3 User's Guide and Reference Manual,}
Technical Report TM-12187, Oak Ridge National Lab., 1993.

\bibitem{HPF} High Performance Fortran Forum,
{\bf High Performance Fortran language specification, version 1.0,}
Technical Report CRPC-TR92225, Center for Research on Parallel Computation,
Rice University, May 1993.

\bibitem{MPI} MPI Forum, ~``Document for a
{\bf Standard Message-Passing Interface},
Technical Report CS-93-214, University of Tennessee, April 1994.

\bibitem{httpd}
The National Center~for~Supercomputing Applications,
The Univ. of~Illinois at~Urbana-Champaign,
{\bf HTTP Daemon},  at URL
{\tt http://hoohoo.ncsa.uiuc.edu/docs/}. (Version current on July 1996).

\bibitem{PCRC} X.~Li, K.~Dincer, and D.~Leskiw,
{\bf Porting NPAC F90D/HPF Compiler onto MPI Platform},
NPAC PCRC Group Internal Report, 1995.

\bibitem{CGI} NCSA,
{\bf Common Gateway Interface}, at URL
{\tt http://hoohoo.ncsa.uiuc.edu/cgi/}. (Version current on July 1996).

\bibitem{CWC} A.S.~Grimshaw, A.~Nguyen-Tuong, W.~A.~Wulf.
{\bf Campus-Wide Computing: Early Results Using Legion at the University
of Virginia,}
Department of Computer Science Technical Report, University of Virginia.

\bibitem{Durra-93}
M.~R.~Barbacci, C.~B.~Weinstock, D.~L.~Doubleday, M.~J.~Gardner, and 
R.~W.~Lichota,
{\bf Durra: a structure description language for developing distributed
applications,}
Software Engineering Journal, pp.83--94, March 1993.

\bibitem{VCE-95}
P.~Rousselle, R.~Yadav, S.~Hariri, D.~Kim, S.~Boddapati, and P.~Tymann,
{\bf Application Development and Execution in a Computing Environment,}
Technical Report, SCCS-705, NPAC, Syracuse University, 1994.

\bibitem{SSL} Netscape Communications,
{\bf Netscape SSLRef,} at URL \\
{\tt  http://home.netscape.com/newsref/std/sslref.html}.
 (Version current on July 1996).


\bibitem{S-HTTP}  Enterprise Integration Technologies,
{\bf Secure HTTP,} at URL \\
{\tt http://www.eit.com/projects/s-http/}. (Version current on July 1996).

%\bibitem{Unify-2} ftp:cs.msstate.edu pub/reports/feicheng.ps.

\end{thebibliography}

\date{}
  
\title{\large\bf Building a World-Wide Virtual Machine
       Based on Web and HPCC Technologies}
  
\author{
\normalsize
  \begin{tabular}[t]{c@{\extracolsep{8em}}c}
  {\bf Kivanc Dincer} & {\bf Geoffrey C.~Fox} \\
  \end{tabular} \\ 
  {\normalsize Northeast Parallel Architectures Center} \\
  {\normalsize Department of Electrical Engineering and Computer Science} \\
  {\normalsize 111 College Place, Mail Stop 3-217} \\
  {\normalsize Syracuse University} \\
  {\normalsize Syracuse, NY 13244-4100} \\
  {\normalsize \tt \{dincer, gcf\}@npac.syr.edu }
}
  
\maketitle
  
\thispagestyle{empty}
  
\subsection*{\centering Abstract}
%\vspace*{-3mm}
\begin{quote}
\begin{small}
In this paper we report on a parallel/distributed virtual machine prototype
called World-Wide Virtual Machine (WWVM) that is designed
to attack grand challenge problems beyond the capabilities
of a single supercomputer.
The prototype is based on emerging Web and existing HPCC  technologies
and utilizes the  pool of Web servers on the internet
as a flexible, convenient, and inexpensive metacomputing resource.
World-Wide Web supplies a standard open interface to the world regardless
of the machine type. Through this interface
Web servers can be used either as computation nodes or coordinators
managing other connected nodes of the virtual machine.
We have constructed a Web-based parallel/distributed programming environment
for solving metaproblems consisting of MPI and PVM  message-passing 
programs and High Performance Fortran programs. \\

{\bf Keywords:} WWW, HPCC, HPF, MPI, PVM, metacomputing, Perl.
\end{small}
\end{quote}

\pagebreak 

\renewcommand{\baselinestretch}{1.5} \normalsize 

\section{Introduction}

Recent advances in both network bandwidths and microprocessor
performance are radically altering high-performance computing environments.
There is a strong trend toward building virtual computers from heterogeneous 
resources distributed on a  network. 
Such virtual computers with enormous computing power and storage capacities
can be used to attack many computationally challenging problems whose
computing and storage requirements are beyond the
capacity of a single dedicated parallel (or super) computer. The virtual 
computer itself may contain different types of architectures and is very 
suitable for solving  
{\bf metaproblems} consisting of many constituent subproblems, each of which 
is suitable for a specific type of computer architecture, yet the metaproblem 
as a whole is outside the scope of a single computer architecture.
It is important to run
 the load in the most suitable computer for a particular problem
in order to use such  an  environment as efficiently as possible.

The existing and emerging World-Wide Web (WWW) infrastructure is expected to
make great contributions toward building a virtual computing environment
that will allow large-scale distributed and high-performance
computing and communications (HPCC) applications to effectively
utilize the computing potential of the existing pool of network-connected
resources. The WWW brings to every type of machine
in the world a standard open interface through which we can manipulate 
individual machines on a network. 
Web servers can be easily  converted to computation nodes
or coordinators mastering other nodes  by using Common Gateway Interface 
(CGI) \cite{CGI} extensions. 
The total computational power of such an assembly of machines
is enormous. 

In this paper we describe how we built such a virtual machine prototype,
 called World-wide
Virtual Machine (WWVM), by using the Web and HPCC technologies. 
Modified  Parallel Virtual Machine (PVM) \cite{PVM} daemons, along with 
CGI-extended Web servers, form  the basis of the communication layer, while
Perl scripts  construct the coordination engine of  this virtual computer.
We aim to demonstrate the usefulness of network-based heterogeneous
 parallel and distributed
processing and to provide a shared resource for collaborating researchers on the WWW.

The WWVM can work in two modes: 
In the first mode, WWVM can be configured as a message-passing distributed 
memory computer. Each Web server in the current configuration is 
an attached { computation node} of this virtual machine. 

In the second mode, WWVM works like a large heterogeneous distributed machine,
or a { metacomputer} for solving metaproblems.
The nodes of this  metacomputer 
can be a single supercomputer, a workstation farm, or even another WWVM 
configured as a message-passing machine. The site taking care of the 
interaction with the user becomes the  {\bf metaproblem coordinator} and  
divides  the problem among other coordinator Web servers running 
on each  specific architecture type. 
The coordinator Web servers are master sites of a specific {\bf virtual 
machine view} and may divide the work into smaller pieces 
according to the availability of
resources under their control. They fetch the inputs for the tasks
(subproblems) assigned to them and notify the metaproblem 
coordinator when they are done. In Section 4.2, we will discuss how the 
problem is divided among servers by using a coordination language.

The prototype consists of various types of workstations 
distributed across the Syracuse University campus. 
We prove the success of this prototype system by 
incorporating a Web-based programming environment, demonstrating Message
Passing Interface (MPI) \cite{MPI} and
PVM message-passing programs and High Performance Fortran (HPF) \cite{HPF}
programs.
We use the NPAC F90D/HPF compiler \cite{Bozkus}  and its associated
F90D/HPF runtime libraries \cite{RTLibs} for HPF processing.
High Performance Fortran  is an emerging shared-memory 
programming standard that extends Fortran 90's array constructs with  data
parallel constructs, intrinsic functions, and data distribution directives.
PVM and MPI are message-passing libraries that have become { de facto}
standards.

Section 2 summarizes the basic layers of the WWVM's system architecture.
Sections 3 and 4 address the Web interface and coordination layer functions, 
respectively.
Section 5 explains how the Web servers are combined to form a virtual computer.
We conclude by commenting on the system  security issues and
briefly describing the important aspects of our work.


\section{WWVM System Architecture}

Our design has a modular structure with three layers: 
(Figure~\ref{fig:layers}).
\begin{itemize}
\item Web interface layer,
\item Configuration and coordination layer,
\item Virtual machine layer. 
\end{itemize}

\begin{figure}[htb]
\centerline{\psfig{file=world-map.eps,width=2.0in}}
\centerline{\psfig{file=wwvm-layers.eps,width=2.0in}}
\centerline{\psfig{file=world-map.eps,width=2.0in}}
\caption{\sf System Architecture of the WWVM.}
\label{fig:layers}
\end{figure}

The {\bf Web interface layer} accepts service requests from the user,
some of which are compilation and execution requests for the
programs supplied through an editor embedded into the Web interface.
The others are related to the
selection and configuration of the computational resources for the specific 
type of problem on hand. When a metaproblem is to be defined,
data and control flow of the meta problem are also defined through this
interface.

The {\bf configuration and coordination layer} serves as a 
processing engine that analyzes the data supplied through the Web interface
and takes necessary actions to set up a { virtual machine view}, which
is a specific virtual machine configuration 
setup by the user by connecting selected Web servers on the internet.
This layer saves the application data and control flow information 
and  coordinates the working of Web compute servers.
This layer is also responsible for starting up tasks and transferring
the data and status  information between the source and destination sites. 

The {\bf virtual machine layer} combines the  Web  servers on the
internet or other machines coordinated by those servers 
to form a parallel/distributed virtual computer from resources on a
geographically distributed network.

\section{Web Interface Layer}

The Web interface provides a metacomputing programming environment 
for the WWVM that allows the user to specify tasks and the dependencies and 
input/output relationships between them.  Individual tasks may be 
written in HPF, Fortran, or C using MPI or PVM message-passing libraries.

\begin{figure}[htb]
\centerline{\psfig{file=web-interface.eps,width=3.5in}}
\caption{Web interface for the WWVM programming environment.}
\label{fig:web-interface}
\end{figure}


Using Netscape or another Web browser of choice, the WWVM user contacts
 the  WWVM Web 
server and is presented with a fill-out form as shown in 
Figure \ref{fig:web-interface}. He/she manipulates push 
buttons, check boxes and text areas
to designate the type of services (HPF, Fortran, or C compilation,
and serial or parallel execution) and the platform (or a WWVM configuration
view) where those services are to be performed, 
and task and data flow information that defines a metaproblem.

The user also has configuration options such as adding or deleting new nodes 
and  forming his own virtual machine view.
The user browses through the Web and
chooses the Web sites volunteering to become part of the virtual machine. 
In fact, in
many cases he is restricted to visiting  only the Web sites
he is known to be collaborating with. If a site is volunteered
to join to configuration, then any  trusted party should be able to use 
the host with a Web password.

After the user has specified the resources he needs through the supplied Web 
form, this information is parsed and analyzed by the interface layer modules,
and configuration parameters are passed to the coordination layer
along with the user's programs and task description information.
The interface layer is also responsible for displaying outputs to the user.

\section{Configuration and Coordination Layer}

The configuration and coordination layer has three subsystems:
configuration, coordination, and compilation which
we will explore  in detail in the following paragraphs.

\subsection{Configuration Subsystem}

The {\bf configuration subsystem} is responsible for configuring
a virtual machine view based on the given configuration parameters 
passed from the interface layer.
It communicates with coordinators or master sites of each virtual machine
view and instructs them to form a virtual machine with the other chosen
sites or machines. We discuss this in detail  in the
next section.

\subsection{Coordination Subsystem}

The metaproblem coordination subsystem maps the individual tasks of 
a complex metaproblem onto  different virtual machine views of WWVM
and manages 
overall coordination and synchronization of tasks executed on each
view. Another important task of this subsystem is to
prepare the environment for the compilation subsystem by transferring required
files to all the nodes before compilation and execution.

We use a coordination language
called {\bf Data Flow/Task Dependency Description Language} that 
helps the programmer describe the main 
tasks and  task groups of the problem and the interaction
between these components to the  coordination subsystem.

Each {\bf task} is an independent computation entity that reads its inputs from
a data file (from the user through the Web-interface or from the output
of another task) and writes its outputs to another data file or to the
Web-based output console.
When  closely correlated tasks with similar platform
requirements are brought together, they form a {\bf task group.} 
These tasks can be launched on the same virtual machine view and can  
therefore--most of the time--can use faster means of data and
message-passing mechanisms than the tasks that were scheduled on different 
nodes.

\begin{figure}[htb]
\centerline{\psfig{file=cl.eps,width=3.00in}}
\caption{\sf Definition of a problem and its task flow graph.}
\label{fig:dfg}
\end{figure}

For each task group we assume that the inputs from other nodes are read in
at once before beginning to execute its tasks, and that all its outputs are 
saved back into the output files when it is finished. 

The tasks are described in the following way using the Web interface: 

\SS
{\scriptsize \sf
\begin{tabbing}
xxxxxxxxxxxx\=xxxxxxx\=xxxxxxxxx\=xxxxxxxxxx\kill
\>\>        \lang TASK = "T1"               \\
\>\>\>           EXECUTABLE = "task1.exe"      \\
\>\>\>           SOURCE = ``simpleGE.hpf :: HPF''  \\
\>\>\>           ARCHITECTURES = ``SUN4'',``ALPHA'' \\ 
\>\>\>           INPUT = ``matrix.dat :: http://npac.syr.edu:8080/task1.in'' \\
\>\>\>           OUTPUT= "reverse.dat::task1.out1'', ``stdout::task1.out2''\\
\>\>        \rang                             \\
\end{tabbing}
}
\DS

  The {\sc task} keyword describes a task named {\sc t1} that has an executable
called {\tt task1.exe} on {\sc sun4} and {\sc alpha} platforms. 
Heterogeneous systems, or systems that do not  share the same file
system, interpret this {\sc architectures} statement and  handles transfer  of
the source codes to the appropriate processing nodes before compilation
or execution.
The executable codes between compatible machines can be directly transferred
between servers. We can 
easily identify that the source file, {\tt simpleGE.hpf}, is written in HPF.
In order 
to execute it, a file called {\tt task1.in} should be ready, and that saves 
its  outputs into {\tt task1.out1} and {\tt task1.out2} once it is executed.

Once all tasks are defined, the user may organize them into task groups.
 For example, the second task 
group (Figure~\ref{fig:dfg}) is defined as:

\SS
{\scriptsize \sf 
\begin{tabbing}
xxxxxxxxxxxx\=xxx\=xxxxxxxxx\=xxxxxxxxxx\kill
\>\>    \lang TASK\_GROUP = "TG\_2"             \\
\>\>\>        CONTAINS = "T5", ``T6'', ``T7''   \\
\>\>    \rang    
\end{tabbing}
}
\DS

Once the task and task group descriptions are parsed and analyzed, an
acyclic data flow/task dependence graph similar to the one in
Figure~\ref{fig:dfg}  is constructed. This tree, augmented
with platform specifications, will later be distributed to the
coordinator nodes of the virtual machine views. The coordinator Web servers
launch a task or task group once all of its inputs are ready after being
notified by the metaproblem coordinator.
In response they inform the metaproblem coordinator about their 
normal or abnormal termination. Appropriate
actions can be taken, depending on this status. For example, there may be
another attempt when failure has occurred or results  may be conveyed  to the 
application layer.

Pulling the data from appropriate tasks is done in a distributed
manner. Each task gets its own data file from the supplying task's server 
location
itself, either using {\sc url\_get} if it is on a remote server, or
using  local copy or symbolic link mechanisms if it is on a local file system. 
We assume that the data is passed to other nodes on different machines by 
writing it to directories that can be accessed by WWW servers and reading
it using Web-based access mechanisms on the other sites. 

Platform specification for the task groups is kept separate from
the task group specifications for flexibility. Tasks are mapped onto 
different views of the WWVM:

\SS
{\scriptsize \sf
\begin{tabbing}
xxxxxxxxxxxx\=xxx\=xxxxxxxxx\=xxxxxxxxxx\kill
\>\>    \lang VIEW = "VM\_1"      \\
\>\>\>        TASK\_GROUP = "TG\_1"   \\
\>\>    \rang                         \\
\>\>    \lang VIEW = "VM\_2"      \\
\>\>\>        TASK\_GROUP = "TG\_2"   \\
\>\>    \rang                         \\
\>\>    \lang VIEW = "VM\_3"      \\
\>\>\>        TASK\_GROUP = "TG\_3"   \\
\>\>    \rang                         \\
\>\>    \lang VIEW = "VM\_1"      \\
\>\>\>        TASK\_GROUP = "TG\_4"   \\
\>\>    \rang                         
\end{tabbing}
}
\DS

In the example, after the node executing {\sc t1} is done, 
the metaproblem coordinator is notified of its completion.
The metaproblem coordinator
activates the next task group by sending a \lang {\sc start} \rang
 message to that
group's master node when all the tasks  supplying input to that 
task group are completed; that is, once task groups {\sc tg\_2} and 
{\sc tg\_3} end, {\sc tg\_4} is fired.

\subsection{The Compilation Subsystem}

The language used in writing  the task's code, the type of the message-passing
libraries used (if any), and the virtual machine view that will be used to 
launch this task are all derived from the task description information
supplied through the Web interface. 

The compilation subsystem is able to compile C, Fortran, and HPF programs 
that are tasks of a metaproblem. It activates a Perl script 
to carry out the requested compile/link/load/execute cycle and to present the
results.  The user can then alter the input program (to correct bugs,
perhaps),
select a new code, or change the service options and may resubmit a new 
form for recompilation.

For compiling the HPF programs  we used NPAC's experimental HPF compiler,
which  translates HPF source code into Fortran 77 with calls
to the message-passing and runtime support (RTS) libraries. The RTS libraries,
which are written using MPI,  provide support for common collective
communication operations  and intrinsic functions of HPF. 


Programs using PVM libraries can run directly on top of the WWVM platform.
On the other hand, the ones written using MPI libraries must be processed
before being executed on this platform.
We wrote an MPI-to-PVM interface layer that emulates MPI calls in terms of PVM 
calls. This interface is based on Mississippi State University's public domain 
software,  Unify \cite{Unify}, which includes  a subset of MPI functions
implemented on top of 
PVM  (version 3.3.2).
PVM functions were modified to add the communication ``contexts'' 
 needed for MPI's protocols. It consists of simple point-to-point
and collective communication, communicator management, and environment
management functions. Both C and Fortran wrappers are provided. 
The system is supported only for Sun and SGI workstations. Since
our RTS libraries were written using MPI,  we implemented functions that
are necessary to run any HPF or MPI program on top of this interface.
One of our important extensions to Unify is the implementation of
routines using cartesian coordinates processor topology. We also ported
Unify to other systems that will be part of the WWVM. 

The compiled codes are linked with the runtime and message-passing 
libraries to obtain executable codes
on each machine. The executables are then mapped onto the nodes of the 
virtual machine view,  with one or more processes per node
according to the number  of machines coordinated by each coordinator.

\section{Virtual Machine Layer}
%-------------------------------
The virtual machine layer combines resources comprised of heterogeneous
workstations (SUN, DEC, SGI, IBM) and small parallel machines (SP-2, CM-5).
%on an irregular (i.e., orders of magnitude differences in bandwidth and latency) but high performance interconnection network.
 We will discuss its main building blocks for the virtual machine layer
below.

\subsection{Communication Daemons}

Communication daemons are responsible for low-level message-passing among
the Web computation nodes. Once a daemon is started on a host it stays alive
until that host is removed from the machine configuration.
 
%{\bf Motiv 1:}
PVM is a { de facto} message-passing parallel processing system that has 
gained wide-spread acceptance in the user community.
We chose to adapt PVM daemons to our environment instead of writing new communication daemons in order to eliminate the 
long initiation times and uncertainty associated with starting from scratch. 

In PVM each user has his own daemons on each node of the virtual machine.
These daemons coordinate communication, spawn processes on 
the same machine, and detect and report host failures. 
A node table containing housekeeping information about each node of the 
virtual machine is compiled and maintained by the master daemon
and is distributed to all other daemons at start-up time.


 
There are further advantages coming directly from using PVM daemons. Some of
these follow:
\begin{itemize}
\item {Fault Detection} \\
An important challenge was to deal with host and network failures in a graceful fashion.
In PVM, a {\bf host failure} is detected if an acknowledgment for a request 
to that host's PVM daemon is not returned.
If a daemon cannot reach its master, it will exit.
When a host failure is detected, it is deleted from host tables, and all
pending or future requests to that host return an error. 
On the other hand, the  {\bf application fault tolerance} has not yet been 
ensured.  While the system itself is 
fault tolerant, the applications block when there is a failure in the system 
and must be manually killed and restarted. 

\item{Data Coercion} \\
Different machines store data in different representations or 
byte orders. 
Messages are converted to a standard network format before sending
and then converted to the  new local format after receiving. 
No conversion is made if the transfer is between two machines of the same
architecture.

\end{itemize}

We modified the PVM daemon structure in various ways to fit into our framework
without affecting the interface seen from outside. This gave us the ability 
to use existing codes written using PVM without change.

%{\bf Modif 1:}
In native PVM, while adding a new host, the PVM daemon at the local host 
activates a PVM daemon on each new host by using ``rsh'' or ``rexec'' 
utilities.  This 
requires a user account and valid password on each new host.
However, when there is a single host that starts remote shells executing
 on other hosts, 
the number of open files quickly exceeds the limits of the operating system
or
simultaneous execution of large number of copies of the same executable file
 may overload the file server causing multiple
lost requests. 
In our Web-based computing environment the user has no
 real accounts (therefore no UNIX passwords) on each machine that is
 part of
the virtual machine. He/she accesses the whole system through  the Web
 site that is responsible for  the user interface. 

\subsection{CGI Configuration Items}

The following 
 are the primary CGI configuration items of the virtual machine
layer. 

\subsubsection{Virtual Machine Console (Front-end)}

The console is located at the master site, i.e., the server site that is   
directly interfaced with the metaproblem coordinator.
The console is used to add or delete nodes to/from the virtual machine 
configuration
or to monitor the general configuration of the machine (see 
Figure \ref{fig:add}).
 When a new node is to
be added, the communication daemons at that node are started by the console. 
The console has three main functions while configuring a virtual machine:

\begin{figure}[htb]
\centerline{\psfig{file=config-vm.eps,width=6.50in}}
\caption{\sf Required steps for adding new servers to the virtual machine.}
\label{fig:add}
\end{figure}

\begin{itemize}
\item { Adding new nodes.}

When a new node is going to be added to the configuration, the console
sends an \lang{\sc add node}\rang~command to the local communication daemon 
at the master server site (Figure \ref{fig:add}).
This module  takes the configuration parameters of the master server daemon
(steps 1-3), 
opens an HTTP connection to the slave site's Web server (step 4), starts its 
slave manager and passes the configuration parameters to it. 
Then it waits until  the slave node's configuration parameters arrive which
it 
passes  to its own daemon (step 8).
This completes the \lang{\sc add node}\rang~operation. 


\item { Deleting nodes.}

The user can remove a node from the virtual machine configuration.
The console instructs its own communication daemon to delete the 
specified node from the virtual machine by removing its address from the node 
table. The new table is broadcast to all other nodes.

\item { Displaying the current machine configuration.}

The console obtains the node table from the local communication daemon 
and displays the current virtual machine
configuration. This configuration  contains information such as
machine architecture types and relative speed of each machine.

\end{itemize}

\subsubsection{Slave Manager}

The slave manager accepts incoming requests from the master site
(or from other slave sites) and takes the necessary actions.
If the incoming request is an \lang{\sc add node}\rang command, then the
request line parameters (Web equivalent of C command line arguments)
include the master site's communication daemon's configuration. 
The slave manager starts a local communication daemon and passes the incoming 
parameters to it (step 5 of Figure \ref{fig:add}). 
In response, the daemon sends him  its own configuration parameters (Step 6),
which are forwarded to the master site as an HTML document (Step 7).

The communication daemon is killed if no acknowledgment is received from 
the master site  daemon within a certain amount of time.
This timeout feature is useful in detecting link or node failures.

%may be the result of a link failure or a host failure.

\subsection{Core Modules}

The following { modules} are used by the master and slave managers to
construct a virtual machine. 

\begin{itemize}
\item Network Module \\
The network module handles all the network access, i.e., 
reading and writing to and
from the network.
A socket-based TCP/IP communication mechanism among servers was
established to exchange messages in HTTP format.                  
This socket connection is fast, reliable,  and it preserves the order of 
messages.

\item Socket Access Module \\
The socket access module controls all the functionality for reading and 
writing  information
from/to a socket or the local file system. This module plays a
great role in exchanging data and commands between a CGI module and a
communication daemon on the same machine.


%{EXTRA:}This module is also used for data and control flow operations and for
%file transfer.

\item Site Access Control Module \\
Hooks for access control (authorization) are provided. A simple filter
based on a Web collaboration password and IP address of the remote host
checks  for the appropriate authorization.
%The authentication system provides the username of the authenticated
%user to the Access Control system.

\item {Protocol Management Module} \\
The protocol management module is invoked in 
order to access a document not found locally. The manager provides
a set of methods handling HTTP or FTP protocol connections
to a remote server
and extracts information using a specific access method.
For example, we can use methods like {\sc url\_get()} and {\sc url\_put()}
to access a document from a remote server using HTTP protocol. 
Data decoding is also done at this stage if the incoming data has been 
encoded before being sent to the local server.

\item {Error Manager} \\
This module keeps track of the errors that occurred during communication 
with a remote server or with a local communication daemon and provides error
messages by filtering the impact of the error on the current status.

\end{itemize}

\section{Security Issues}

As always, it was a
challenge  to provide access to more powerful, higher-level computing
services without compromising security. Although  we
used Web authentication based on Web collaboration passwords and IP addresses,
this mechanism may not be considered secure since the user-id's and passwords
are transmitted on the internet in ``plain-text'' form, allowing  intruders
listen to lines and steal legitimate users' passwords. We precluded some
possible mischief  in such a case by powerfully restricting the 
abilities of whatever user or
pseudo user the HTTP daemon runs (for example, user ``nobody'') and by
monitoring the access records carefully for signs of abuse \cite{Cowie}.

Several companies are working on providing secure Web  protocols and systems 
that can be used in commercial applications. There are currently two protocols available for conducting a secure
transaction. The first  is Secure Socket Layers (SSL) \cite{SSL},
by Netscape Communications. The other is Secure-HTTP (S-HTTP) \cite{S-HTTP},
proposed by EIT. We have already begun seeing 
servers and browsers supporting secure transactions. We believe that once 
secure means of  encryption and authentication become widespread, Web 
security issues will converge to UNIX security issues.

\section{Summary}

We demonstrated the feasibility of using Web technologies combined with
HPCC tools to construct a heterogeneous parallel/distributed virtual machine. 
We successfully integrated 
a Web-based metaproblem programming environment on top of this platform
that enables definition and execution of metaproblems consisting of HPF,
Fortran, or C tasks using PVM and MPI message-passing libraries.

\subsection*{Acknowledgments}

We would like to thank Wojtek Furmanski, Tom Haupt, Donald Leskiw, and
Xiaoming
Li for providing their feedback in various stages of this work, and Elaine
Weinman for proofreading this manuscript.

\renewcommand{\baselinestretch}{1}
\begin{small}

\end{small}

\section*{Author Biographies}

\hspace{0.2in}
{\bf Kivanc Dincer} is a Ph.D. candidate at 
Syracuse University and a  research 
assistant at Northeast Parallel Architectures Center. He has an M.S. degree
in Computer Science from Iowa State University. He worked on the
design and development of several HPCC-related research projects including 
NPAC F90D/HPF compiler and Parallel Compiler Runtime Consortium runtime 
support systems development projects. His major research interests
include parallel processing, distributed computing and Web-based parallel 
programming environments. 
({\bf dincer@npac.syr.edu}) \\
({\bf http://www.npac.syr.edu/users/dincer/}) \\

{\bf Geoffrey C.~Fox}
is an internationally recognized expert in the use of parallel
architectures and the development of concurrent algorithms.  He leads a
major project to develop prototype high performance Fortran
(Fortran90D) compilers.  He is also a leading proponent for the
development of computational science as an academic discipline and a
scientific method.  His research on parallel computing has focused on
development and use of this technology to solve large scale
computational problems.  Fox directs InfoMall, which is focused on
accelerating the introduction of high speed communications and parallel
computing into New York State industry and developing the corresponding
software and systems industry. Much of this activity is centered on
NYNET with ISDN and ATM connectivity throughout the state including
schools where Fox is leading developments of new K-12 applications that
exploit modern technology. 
({\bf gcf@npac.syr.edu}) ({\bf http://www.npac.syr.edu/users/gcf/})

\end{document}

  

